Author's response to reviews

Title: Characterization of a new full length TMPRSS3 isoform and identification of mutant alleles responsible for nonsyndromic recessive deafness in Newfoundland and Pakistan

Authors:

Zubair M Ahmed (ahmedz@nidcd.nih.gov)
Xiaoyan CINDY Li (CLI@hei.org)
Shontell D Powell (spowell@law.gwu.edu)
Saima Riazuddin (riazuddin@nidcd.nih.gov)
Terry-Lynn Young (tlyoung@mun.ca)
Khushnooda Ramzan (camb1@wol.net.pk)
Zahoor Ahmad (camb1@wol.net.pk)
Sandra Luscombe (Sandra.Luscomb@hccsj.nl.ca)
Kiran Dhillon (kkd@u.washington.edu)
Linda MacLaren (Linda.MacLaren@CalgaryHealthRegion.ca)
Barbara Ploplis (ploplisb@nidcd.nih.gov)
Lawrance I Shotland (Larry.Shotland@med.va.gov)
Elizabeth Ives (Elizabeth.Ives@caphealth.org)
Sheikh Riazuddin (riaz@lhr.comsats.net.pk)
Thomas B Friedman (friedman@nidcd.nih.gov)
Robert J Morell (morellr@nidcd.nih.gov)
Edward R Wilcox (wilcoxe@nidcd.nih.gov)

Version: 5 Date: 22 September 2004

Author's response to reviews:

In response to your latest Email:

"One of the errors was not entirely corrected:

Section headings - Please remove all unnecessary capitalisation and fullstops from all headings and sub-headings in the manuscript. This causes problems during the xml markup stage.

I have edited your manuscript and uploaded the new version. Please let me know if you are unhappy with the changes."

I have tried to make sure all formatting problems are corrected, please tell me if I have missed something. Also, Lawrence Shotland was not happy with the audiology paragraph and has added another sentence with regards to how the audiological assessment was performed.